Author's response to reviews

Title: A cross-sectional study of HIV and syphilis infections among male students who have sex with men (MSM) in northeast China: implications for implementing HIV screening and intervention programs

Authors:

Jun-jie Xu (xijbeijing@gmail.com)
Kathleen H Reilly (ktreilly.ncaids@gmail.com)
Chun-Ming Lu (chunminglu@126.com)
Ning Ma (maning_m2000@163.com)
Min Zhang (minzhang_100@hotmail.com)
Zhen-Xing Chu (laochu323@yahoo.com.cn)
Jun-jie Wang (jjw405047@sohu.com)
Ke Yun (yunke032433@gmail.com)
Hong Shang (hongshang100@hotmail.com)

Version: 3 Date: 16 April 2011

Author's response to reviews:

The address of the first co-author and corresponding author was slightly changed from

"Key Laboratory of AIDS Immunology of Ministry of Health, No 1 Hospital of China Medical University, Shenyang, China" to

"From Key Laboratory of AIDS Immunology of Ministry of Health, Department of Laboratory Medicine#No.1 Hospital of China Medical University, Shenyang 110001, China"

The statistical tables in the manuscript were changed, in which the border lines of statistical tables were hided.